Author's response to reviews

Title: Protocol for the 'e-Nudge Trial': A randomised controlled trial of electronic feedback to reduce the cardiovascular risk of individuals in general practice [ISRCTN64828380]

Authors:

Tim A Holt (tim.holt@warwick.ac.uk)
Margaret Thorogood (margaret.thorogood@warwick.ac.uk)
Frances Griffiths (f.e.griffiths@warwick.ac.uk)
Stephen Munday (stephen.munday@SwarkPCT.nhs.uk)

Version: 2 Date: 25 April 2006

Author’s response to reviews:

We have made revisions as instructed to improve the formatting. No other changes have been made.